Author's response to reviews

Title: Sexual and reproductive health beliefs and practices of female immigrants in Spain. A qualitative study

Authors:

Carmen Álvarez-Nieto (calvarez@ujaen.es)
Guadalupe Pastor-Moreno (gpastormoreno@yahoo.es)
Mª Luisa Grande-Gascón (mlgrande@ujaen.es)
Manuel Linares-Abad (mlinares@ujaen.es)

Version: 6 Date: 14 August 2015

Author's response to reviews: see over
14 August 2015.

To the Editor in chief of Reproductive Health

Dear Prof. Bloom,

Many thanks for revising our manuscript. We have checked the contents of Table 1 and we have corrected the question.

We regret the error.

Thank you very much again.

Yours sincerely.

On behalf of all authors,

Prof. Carmen Alvarez Nieto